Author’s response to reviews

Title: The Unveiling of a New Risk Factor Associated with Bladder Cancer in Lebanon

Authors:

Sally Temraz (st29@aub.edu.lb)
Yolla Haibe (yh25@aub.edu.lb)
Maya Charafeddine (mc16@aub.edu.lb)
Omran Saifi (oas12@mail.aub.edu)
Deborah Mukherji (dm25@aub.edu.lb)
Ali Shamseddine (as04@aub.edu.lb)

Version: 3  Date: 10 Feb 2019

Author’s response to reviews:

1. Ethics and Consent to Participate

Please can you clearly state ethical approval and consent were not required due to the retrospective nature of the study.

Response: Thank you, we clearly stated in the ethical approval and consent section that they were not required due to the retrospective nature of the study.

2. Availability of Data and Materials

The Availability of data and materials section refers to the raw data used in your study and presenting tables and figures is not sufficient to state that all data is contained within the manuscript and additional files. Please only use this statement if you have indeed provided all raw data on which your study is based.

- The datasets generated and/or analysed during the current study are available in the [NAME] repository, [PERSISTENT WEB LINK TO DATASETS]
• The datasets used and/or analysed during the current study available from the corresponding author on reasonable request.

• All data generated or analysed during this study are included in this published article [and its supplementary information files].

• The datasets generated and/or analysed during the current study are not publicly available due [REASON WHY DATA ARE NOT PUBLIC] but are available from the corresponding author on reasonable request.

• Data sharing is not applicable to this article as no datasets were generated or analysed during the current study.

• The data that support the findings of this study are available from [third party name] but restrictions apply to the availability of these data, which were used under license for the current study, and so are not publicly available. Data are however available from the authors upon reasonable request and with permission of [third party name].

• Not applicable. If your manuscript does not contain any data, please state 'Not applicable' in this section.

Response: Thank you, we clearly stated that the datasets used and/or analysed during the current study available from the corresponding author on reasonable request.

3. Authors Contributions

Please include a statement in the Authors' contributions section to the effect that all authors have read and approved the manuscript, and ensure that this is the case.

Thank you, we have included the sentence that all authors have read and approved the final manuscript.

4. Clean Manuscript

At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Please ensure that all figures, tables and additional/supplementary files are cited within the text.

Thank you, we have uploaded the manuscript as clean version without any tracked changes or comments.